ഹ



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2021GL095136

Key Points:

- This study quantifies, for the first time, precipitation recycling ratios over the Amazon and Congo Basins in Coupled Model Intercomparison Project Phase 6 models
- Seasonal recycling is well represented in the Congo but there is a large-scale underestimation in the Amazon dry-to-wet transition season
- These findings have implications for model representation of land-use change impacts in the Amazon

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. C. A. Baker, j.c.baker@leeds.ac.uk

Citation:

Baker, J. C. A., & Spracklen, D. V. (2022). Divergent representation of precipitation recycling in the Amazon and the Congo in CMIP6 models. *Geophysical Research Letters*, 49, e2021GL095136. https://doi.org/10.1029/2021GL095136

Received 12 JUL 2021 Accepted 17 APR 2022

Author Contributions:

Conceptualization: J. C. A. Baker, D. V. Spracklen Formal analysis: J. C. A. Baker Funding acquisition: D. V. Spracklen Methodology: J. C. A. Baker Project Administration: D. V. Spracklen Supervision: D. V. Spracklen Writing – original draft: J. C. A. Baker Writing – review & editing: J. C. A. Baker, D. V. Spracklen

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Divergent Representation of Precipitation Recycling in the Amazon and the Congo in CMIP6 Models

J. C. A. Baker¹ D and D. V. Spracklen¹

¹School of Earth and Environment, University of Leeds, Leeds, UK

Abstract Moisture evaporated from the land contributing to precipitation in a given area is known as precipitation recycling and needs to be accurately represented in climate models. The Amazon and Congo basins are reported to have the highest precipitation recycling rates globally, but model representation has not yet been assessed over these regions. We evaluated recycling over the Amazon and Congo in 45 Coupled Model Intercomparison Project Phase 6 models. Regional annual means from models and reanalyzes agreed well over both basins. Models captured seasonal variation in recycling over the Congo but there was a large-scale underestimation of recycling during the Amazon dry-to-wet transition season relative to ERA5, caused by models underestimating Amazon evapotranspiration and overestimating incoming wind speed and associated water vapor imports. Both regions show robust declines in precipitation recycling over the next century under future climate-change scenarios. Our results suggest models may underestimate impacts of deforestation on regional precipitation in the Amazon.

Plain Language Summary In the Amazon and Congo basins, much of the rain that falls to Earth is transferred back to the atmosphere through evaporation over tropical forests, providing essential moisture for further rainfall events downwind. This process of rainfall recycling, which is vital for sustaining the rainforest, is disrupted when forest is cut down. For researchers to understand how deforestation in the Amazon and Congo might impact the climate it is vital that climate models accurately represent the fraction of rain coming from recycled moisture. Our study found that models represented rainfall recycling relatively well in the Congo, but underestimate recycling in the Amazon dry-to-wet transition season. This is important because it could mean these models underpredict the influence of deforestation on climate in the Amazon.

1. Introduction

Moisture for precipitation over a particular area can either originate from evapotranspiration (E) within the area, or from water vapor transported from elsewhere (Brubaker et al., 1993; Eltahir & Bras, 1994; Gimeno et al., 2012; Trenberth, 1999). The ratio of locally-derived to total precipitation is known as the precipitation recycling ratio and provides a method of quantifying the land-surface contribution to regional precipitation. Areas with high precipitation recycling are expected to show greater climate sensitivity to land-use change, such as tropical deforestation, due to the effects on surface water fluxes and downwind precipitation (Eltahir & Bras, 1996; Spracklen et al., 2012). In order to accurately simulate the regional climate impacts of land-use change, climate models must provide an accurate representation of precipitation recycling in areas where: (a) precipitation recycling is known to be important, and (b) where deforestation rates are high.

The Amazon and Congo river basins are reported to have the highest precipitation recycling rates globally (Tuinenburg et al., 2020). These regions contain extensive areas of tropical forest (Saatchi et al., 2011), characterized by high rates of E, which, combined with prevailing winds, allow moisture to be propagated thousands of kilometers inland (Salati et al., 1979; Spracklen et al., 2018; Staal et al., 2018; van der Ent et al., 2010). The Amazon hydrological cycle is characterized by a single wet season with precipitation peaking during the austral summer (December–February, DJF, Marengo, 2006), while the Congo basin has two rainy seasons, with peaks in March and November (Crowhurst et al., 2021). A literature survey for this study (Table S1 in Supporting Information S1) found annual-mean recycling estimates for the Amazon ranging from 24% to 41% (Brubaker et al., 1993; Burde et al., 2006; Costa & Foley, 1999; Eltahir & Bras, 1994; Staal et al., 2018; Trenberth, 1999; Tuinenburg et al., 2020; van der Ent et al., 2010; Yang & Dominguez, 2019; Zemp et al., 2014) compared to 28%–50% for the Congo (Dyer et al., 2017; Pokam et al., 2012; Sorí et al., 2017; Tuinenburg et al., 2020). The spread in recycling

estimates reported for the Amazon and Congo can be attributed to differences in the modeling approaches used, variation in inputs, analysis domain boundaries, and quality of the input data (Zemp et al., 2014).

Rates of forest loss for the Amazon and Congo are among the highest in the world (Hansen et al., 2013; Potapov et al., 2017; Tyukavina et al., 2018). Primary forest loss for the Amazon and Congo from 2002 to 2019 was estimated at 30.5 and 6.1 Mha, respectively, equivalent to 0.31% and 0.19% yr⁻¹ (Butler, 2020; Hansen et al., 2013). Historical and future forest loss is projected to reduce regional precipitation through a reduction in precipitation recycling over the two basins though results vary widely among modeling studies (see Lawrence & Vandecar, 2015; Perugini et al., 2017; Spracklen & Garcia-Carreras, 2015; Spracklen et al., 2018 and references therein). In addition to variation in simulation setup, differences between model land-surface schemes and parameterization of sub-grid scale processes have been suggested to be important in driving these differences (Bell et al., 2015; Crowhurst et al., 2020). However, representation of precipitation recycling over the Amazon and Congo in global climate models has not previously been assessed.

In this study we make a first attempt to assess the representation of regional precipitation recycling over the Amazon and the Coupled Model Intercomparison Project Phase 6 (CMIP6) models. Our results provide insights on the ability of CMIP6 models to represent the land-surface contribution to rainfall over two important regions of tropical forest, with implications for their representation of land-use change impacts on climate.

2. Data and Methods

2.1. CMIP6 Model Output

We analyzed precipitation recycling using output from 45 global climate models contributing to CMIP6 (Eyring et al., 2016; Table S2 in Supporting Information S1). Monthly model output available in March 2021 was downloaded from the World Climate Research Program Earth System Grid Federation. We retrieved historical simulations (1979–2014) and simulations from four future Shared Socioeconomic Pathway (SSP) experiments representing alternative development scenarios over the next century. These can be approximately interpreted as low (SSP1-2.6), medium-low (SSP2-4.5), medium-high (SSP3-7.0) and high (SSP5-8.5) radiative-forcing levels by 2100 (O'Neill et al., 2017). In terms of land-use change, the SSP3-7.0 scenario has the largest reduction in forest cover over the next century, SSP5-8.5 scenarios (Lawrence et al., 2016). We selected models that provided historical simulations of the following variables (Table S3 in Supporting Information S1): three-dimensional specific humidity (q, kg kg⁻¹), zonal (u) and meridional (v) wind fields (m s⁻¹), evapotranspiration (E, kg m⁻² s⁻¹) and surface temperature (K). When available, multiple realizations were used to derive an ensemble mean to remove the influence of random error and uncertainty associated with initial conditions (e.g., Krasnopolsky & Lin, 2012; Palmer et al., 2004).

2.2. Reference Data

Calculation of precipitation recycling requires consistent representation of the atmospheric moisture budget; thus, reanalyzes were used to evaluate model performance (Text S1 and Table S4 in Supporting Information S1). We compared the Amazon and Congo seasonal cycles in E and P from each reanalysis with independent datasets to assess whether the processes used to derive E and P are likely to be realistic (Text S2 in Supporting Information S1). The catchment-balance approach provides data-driven, independent estimates of E (Baker et al., 2021; Burnett et al., 2020), and a merged satellite-rain gauge product (CHIRPS2, Funk et al., 2015) was used to evaluate precipitation. Based on these results, only ERA5 (Hersbach et al., 2020) and JRA-55 (Kobayashi et al., 2015) were used to evaluate CMIP6 models at the seasonal timescale, with ERA5 selected as the primary reference data set since it showed the strongest correlations across both regions (Table S5 in Supporting Information S1). An important caveat is that reanalyzes, including ERA5, do not fully capture the strong seasonality in Amazon E that the catchment-balance E estimates indicate (Baker et al., 2021). However, since we draw consistent conclusions when comparing Amazon CMIP6 E to ERA5 and catchment-balance E, we do not expect the underestimation of Amazon E seasonality in ERA5 to affect the main conclusions that we draw in this study.





Figure 1. Annual mean precipitation (P) recycling estimates for (a) the Amazon and (b) the Congo from 45 Coupled Model Intercomparison Project Phase 6 models (blue boxplots) and four reanalyzes (black markers) calculated in this study using the Brubaker model and the EB94 model (data from 1979 to 2014), compared against values from the literature (green markers). Climatological annual-mean lower-tropospheric (1000–850 hPa) winds from ERA5 (vectors) and the domains used for each basin (green boxes) are shown in panels (c and d). Box plots show the quartiles (box), mean (x marker) and the upper and lower extremes (whiskers). Note that the literature values from Bosilovich and Chern (2006) are for October–December only and shown in a different color (light green circles).

2.3. Precipitation Recycling

Monthly precipitation recycling was estimated for the Amazon $(19^{\circ}S-5^{\circ}N, 79^{\circ}W-50^{\circ}W)$ and Congo $(13^{\circ}S-9^{\circ}N, 12^{\circ}E-34^{\circ}E)$. These domains were chosen to capture the entirety of the river basin in each region (Figures 1c and 1d). To estimate recycling we applied the models from Brubaker et al. (1993) and Eltahir and Bras (1994, henceforth EB94). These are both examples of "bulk" recycling models, and were selected because they are relatively computationally simple (Bosilovich & Schubert, 2002), making them an appropriate choice for a multi-model comparison study. Although we observed some differences in the magnitude of recycling ratios derived from the two approaches, results at the seasonal timescale showed good agreement. For brevity, we describe the EB94 model in this section and provide a description of the Brubaker model in Text S3 in Supporting Information S1.

The EB94 method provides insights on the spatial variation in recycling and has been shown to compare well to more sophisticated models that include water vapor tracers (Dominguez et al., 2022). Calculations are performed at the grid-cell level using an iterative approach to derive the precipitation recycling ratio (ρ). The model requires E plus column-integrated moisture fluxes (qu and qv, kg m⁻¹ s⁻¹), calculated as mass-weighted vertical integrals (Text S4 in Supporting Information S1). Moisture fluxes across the domain are decomposed into moisture that originated from within the domain (i.e., of local origin) and fluxes that originated externally. For each grid cell, ρ is the ratio of moisture outflux of local origin (O_{local}) to total outflux (O), which is equal to the ratio of local precipitation (P_{local}) to total precipitation (P):

$$\frac{O_{\text{local}}}{O} = \frac{P_{\text{local}}}{P} = \rho \tag{1}$$

Rearranging Equation 1:

$$O_{\rm local} = O \times \rho \tag{2}$$

In practice, ρ is initially guessed for each grid cell and used to compute O_{local} . O_{local} for one grid cell is the incoming moisture of local origin (I_{local}) for an adjacent grid cell, allowing ρ values for each cell to be recalculated



iteratively, using the prior ρ estimates of the surrounding grid cells to determine the local moisture fluxes (with total *O* derived directly from *qu* and *qv*):

$$\rho^{(n+1)} = \frac{\rho^n \times O + e}{O + e} = \frac{I_{\text{local}} + e}{I + e}$$
(3)

where *n* is the iteration number and *e* is the evapotranspiration over the grid cell multiplied by the cell area in metres. Eltahir and Bras (1994) showed that recycling values quickly converge using this approach. All datasets were re-gridded to that of the coarsest-resolution model ($2.8125^{\circ} \times 2.8125^{\circ}$) before computing recycling ratios.

The recycling models applied here assume a well-mixed atmosphere, which may not be valid over areas of moist tropical forest. Lettau et al. (1979) described the "fast recycling" that occurs over the diurnal cycle in the Amazon, with local clouds developing and raining out before evaporated moisture has fully mixed through the atmospheric column. The well-mixed assumption could mean that recycling is underestimated. Another caveat is that Equations 1 and 2 were applied to monthly data, neglecting the transient eddy contribution to water vapor fluxes, potentially causing a slight underestimation of recycling values (Text S5 and Figure S1 in Supporting Information S1). In this study, we were primarily interested in benchmarking recycling in models against reanalyzes and found good agreement between our EB94 results and those from the literature (Figure 1).

3. Results and Discussion

Annual-mean ρ values for the Amazon and Congo estimated from models and reanalyzes are compared with literature values in Figure 1. For both recycling approaches, CMIP6 values agree well with reanalyzes, though estimates from the Brubaker method are lower than the EB94 estimates, which are more consistent with previously-reported values. Mean Amazon ρ from the CMIP6 ensemble using the EB94 method is 36.5 ± 4.9% (mean \pm standard deviation), which is very close to the mean of the reanalyzes (36.2 \pm 1.2%, Figure 1a, c.f. 18.7 ± 3.3 vs. 21.6 ± 1.5 from the Brubaker method, Table S6 in Supporting Information S1). In the Congo, annual-mean ρ estimates are higher than in the Amazon, and CMIP6 models and reanalyzes agree relatively well (Figure 1b). The Congo CMIP6 mean from EB94 is $41.9 \pm 4.9\%$ and the reanalysis mean is $39.7 \pm 3.1\%$. Our results confirm the expectation that the Brubaker method may underestimate the true magnitude of recycling in the Amazon and Congo Basins, while the EB94 method is likely to be more accurate. In the Amazon, CMIP6 mean recycling of 18.7 ± 3.3 from the Brubaker method lies well below the mean of literature values $(29.8 \pm 5.0\%)$, Table S1 in Supporting Information S1), while CMIP6 mean recycling of $36.5 \pm 4.9\%$ from the EB94 method is closer to literature values. Similarly, in the Congo, mean CMIP6 recycling of $28.4 \pm 11.4\%$ from the Brubaker method lies well below the mean of literature values ($40 \pm 11\%$), while mean recycling of $41.9 \pm 4.9\%$ from EB94 is more consistent. Given the better performance of EB94 relative to literature values, we focus the rest of our discussion on presenting results from this model and provide results from the Brubaker method in Supporting Information S1.

Spatial variation in ρ over the Amazon and Congo basins in each season is shown in Figure 2. In the Amazon, models capture a gradient in recycling across the basin, with the highest values in the western margins (Figure 2, columns 1 & 2). However, CMIP6 models show a spatially coherent underestimation of Amazon recycling in June–August (JJA, Figure 2o) and September–November (SON, Figure 2u), relative to ERA5. In SON, negative ρ biases are observed across 70% of the basin and by the majority of models (Figure 2u). During DJF and March–May, CMIP6 models simulate higher ρ than ERA5 over the Amazon, with the biases in different parts of the year compensating to result in comparable regional annual means (Figure 1).

In the Congo, recycling is highest in the southeast during DJF (Figure 2d) and in the northern basin in JJA (Figure 2p), and this spatial pattern is well replicated by the models (Figures 2e and 2q). CMIP6 models show a negative bias relative to ERA5 in the central northern Congo in JJA (Figure 2r), though in general models show less agreement in the direction of biases over the Congo compared to the Amazon (less stippling in Figure 2), and biases are less spatially extensive, suggesting that the CMIP6 ensemble mean generally represents recycling better over the Congo than over the Amazon at the seasonal scale.

Climatological seasonal cycles in ρ and its controlling variables for the whole Amazon and Congo are presented in Figure 3. In the Amazon, ERA5 and JRA-55 show annual peaks in ρ in SON of 41.2% and 42.3%, respectively,



Geophysical Research Letters



Figure 2. Spatial variation and absolute differences in ρ in ERA5 and the Coupled Model Intercomparison Project Phase 6 ensemble mean over the Amazon and Congo for (a–f) December–February, (g–l) March–May, (m–r) June–August, and (s–x) September–November. Recycling was estimated for 1979–2014 using EB94, with grid-cell values weighted by normalized precipitation. Stippling indicates where at least 75% of the models agree on the direction of biases relative to ERA5. Relative difference maps are shown in Figure S2 in Supporting Information S1.

compared to 38.8% for the CMIP6 models (Figure 3a; 26.7% and 28.1% vs. 22.4% in Brubaker estimates, Figure S3a in Supporting Information S1), consistent with biases shown in Figure 2u. Less than half (20/45) of CMIP6 models captured a statistically significant positive correlation with the ERA5 Amazon seasonal cycle (Table S7 in Supporting Information S1), with the ensemble-mean showing the highest ρ in March (Figure 3a). The underestimation of E (relative to both ERA5 and catchment-balance) and overestimation of moisture influx (relative to ERA5) both contribute to the CMIP6 underestimation of Amazon ρ in SON (Figures 3b and 3c). The CMIP6 underestimation of recycling relative to ERA5 shown in Figure 3a is likely to be a conservative estimate, since ERA5 does not capture the strong dry-season increase in E indicated by catchment-balance (Figure 3b, Text S2 in Supporting Information S1). Correlations indicate that seasonal variation in Amazon ρ is primarily controlled by E but this is not captured by the models (Table S8 in Supporting Information S1).

The combination of CMIP6 models underestimating E and simulating excess advected moisture results in the land-surface contribution to precipitation being underestimated in SON, which represents the transition between the dry and wet seasons in the Amazon (Figure S4a in Supporting Information S1). Previous studies have shown Amazon deforestation has the greatest impact on climate at the end of the dry season, as forest loss reduces the transfer of deep groundwater to the atmosphere by transpiration (Baker & Spracklen, 2019; da Rocha et al., 2004; Davin & de Noblet-Ducoudré, 2010; von Randow et al., 2004). Furthermore, the underestimation is concentrated





Figure 3. (a and d) Seasonal variation in precipitation (P) recycling estimated using EB94; (b and e) domain-average evapotranspiration (E); and (c and f) moisture influx, over the Amazon and Congo basins calculated using two reanalyzes (black lines) and 45 Coupled Model Intercomparison Project Phase 6 (CMIP6) climate models (narrow blue lines) using data from 1979 to 2014. The CMIP6 ensemble mean and interannual standard deviation are shown by the thick blue line and blue shading, respectively. Independent catchment-balance E estimates for the Amazon (Baker et al., 2021) and the Congo (Burnett et al., 2020) are indicated in green in panels (b and e). The catchment-balance E estimates are from 2003 to 2013 (Amazon) and 2002 to 2016 (Congo). The domains used for each basin are indicated by the green dashed boxes in Figures 1c and 1d.

in the eastern Amazon (Figure 2u), which is important because the southeast of the basin is a deforestation hotspot (Hansen et al., 2013), and it is therefore crucial that models accurately represent land-atmosphere interactions in this region. These results have implications for analyses of deforestation impacts on future precipitation, since in the models a smaller fraction of rainfall is derived from the land surface than in reality. This could result in CMIP6 models underestimating the impacts of Amazon land-use change on climate in future scenarios.

In the Congo, CMIP6 models show relatively good agreement with reanalyzes (Figure 3d), with three quarters (33/45) showing statistically significant positive correlations with ERA5 at the seasonal scale (Table S7 in Supporting Information S1). The spread across reference datasets is higher in the Congo than in the Amazon (Figures 1b and 2), reflecting more-limited in situ observations to assimilate and constrain the reanalyzes over this region (e.g., Haiden et al., 2018). CMIP6 models capture a bimodal distribution in ρ with peaks during the drier parts of the year (JJA and DJF). Results from both recycling approaches were found to be consistent (Figures 3d, S3b, and S4b in Supporting Information S1). The correspondence between E estimates from catchment-balance, reanalyzes and CMIP6 models shown in Figure 3e suggests that the simulated seasonal cycle in Congo E is relatively robust. E, which might be expected to positively correlate with ρ , broadly follows the same seasonal cycle as precipitation in the Congo (Figure S4b in Supporting Information S1), providing essential atmospheric moisture without being the primary driver of precipitation seasonality (Cook & Vizy, 2021). An imbalance in E between the two rainfall peaks has been linked to a difference in solar radiation (Burnett et al., 2020; Crowhurst et al., 2021; Crowhurst et al., 2020). Our results indicate the seasonal cycle of ρ in the Congo is most strongly influenced by variation in moisture influx (Figure 3f), with strong statistically-significant negative correlations between ρ and influx in reanalyzes and CMIP6 models (r = -0.88--0.93, p < 0.001, Table S8 in Supporting Information S1).

To further understand representation of precipitation recycling in CMIP6 models, we decomposed advected moisture into its component fields, that is, wind and specific humidity (Hari et al., 2020). In the Amazon, models overestimate winds in the lower troposphere (1000–850 hPa, Figure S5a in Supporting Information S1), which is the height in the atmosphere with the highest humidity content and most important for moisture transport (Ralph et al., 2005). Since models represent incoming humidity relatively well (Figure S5b in Supporting Information S1), this causes an overestimation of moisture imports to the Amazon. In the Congo, CMIP6 models also overestimate lower troposphere incoming winds (Figure S6a in Supporting Information S1), though a compensating underestimation of incoming humidity (Figure S6b in Supporting Information S1) results in adequate representation of moisture imports (Figure 3f). This shows that the good representation of recycling in CMIP6 models over the Congo (Figure 3d) is partly attributable to compensating errors in underlying model processes.

To understand why CMIP6 models tend to overestimate incoming winds over both study regions, we analyzed associations between basin incoming wind speed and land-surface temperature biases. Wind biases relative to ERA5 are positively correlated with surface temperature biases over each basin (r = 0.52, and r = 0.62, p < 0.001, for Amazon and Congo, respectively, Figure S7 in Supporting Information S1). Positive biases in CMIP6 winds have previously been reported over parts of Africa (Akinsanola et al., 2021), while CMIP6 models underestimate wind speed over the tropical Atlantic (Richter & Tokinaga, 2020). Our analysis of CMIP6 winds focussed on the strength of winds entering the Amazon and Congo, though we might expect differences between conditions over the land and ocean to have some influence over wind biases (Richter et al., 2012). These results suggest why CMIP6 models tend to overestimate moisture advection to the Amazon, illustrating how biases in one variable can result in further biases elsewhere in the climate system.

Comparing results from different versions of the same model or between closely-related models can indicate the effect of model development or resolution increases on model performance. Improvements in model representation of precipitation recycling are more noticeable over the Amazon, as in the Congo most models perform well (Table S7 in Supporting Information S1). In the Amazon, increases in model resolution improved representation of seasonal recycling in four CMIP6 models (CNRM-CM6-1, HadGEM3-GC31, MPI-ESM1-2, NorESM2). Some modeling centers provided multiple model versions, allowing comparisons between models of differing complexity. There are two instances of Earth System models (ESMs), which explicitly represent biogeochemical cycling (Flato, 2011), showing better representation of recycling than the coupled model from the same modeling center (ACCESS-ESM1-5, MIROC-ES2L), and one instance where the ESM performs less well than the equivalent coupled model at the same resolution (CNRM-ESM2-1).

Finally, we analyzed future trends in precipitation recycling over the Amazon and Congo in model simulations from four future climate scenarios (Figure 4). By 2100, recycling is robustly projected to decline over both basins relative to the 1979–2014 baseline, with stronger declines under higher warming scenarios. Reductions in ρ are larger over the Amazon than in the Congo (relative changes of -17.5% and -12.0% by 2100 under the SSP585 scenario, Figures 4a and 4b). The decline in ρ is primarily caused by large increases in imported moisture over both basins (Figures 4e and 4f), while the changes in E are more modest and vary between scenarios (Figures 4c and 4d), likely affected by a combination of land-use change and atmospheric carbon dioxide concentrations (Lawrence et al., 2016). The increase in horizontal moisture transport over the course of the century is consistent with the amplification of the global water cycle in a warming climate (Held & Soden, 2006; Skliris et al., 2016) and was previously observed in the CMIP5 models (Lavers et al., 2015). As moisture transport increases, the land-surface contribution to precipitation is increasingly outweighed by advected moisture.

Our results have implications for projected impacts of deforestation on regional precipitation, as the climate impacts of Amazon land-use change could be underrepresented in CMIP6 models. Figure 4 suggests that climate sensitivity to the land surface may decline in the future. However, it should be noted that due to the "fast recycling" of moisture in the lower boundary layer over the Amazon and Congo basins (Lettau et al., 1979; Pokam





Figure 4. (a and b) Future changes in precipitation (P) recycling calculated using EB94; (c and d) evapotranspiration (E); and (e and f) moisture influx, over the Amazon and Congo basins from four shared socio-economic pathway (SSP) scenarios, calculated relative to the 1979–2014 climatological mean. An 11-year moving average was applied to reduce noise. Each narrow line represents an individual model, and the thick lines show the ensemble means for each SSP.

et al., 2012), future reductions in precipitation recycling caused by increasing moisture transport may not be as extreme as models suggest. Future work evaluating the sources of moisture for precipitation in the presence of incomplete vertical mixing, and assessing how this is represented in climate models would shed further light on the validity of future climate projections.

Data Availability Statement

The Coupled Model Intercomparison Project Phase 6 data set is available at https://esgf-node.llnl.gov/projects/ cmip6. ERA5 data are available at https://doi.org/10.24381/cds.6860a573 and https://doi.org/10.24381/cds. f17050d7. JRA-55 data are available at https://doi.org/10.5065/D60G3H5B. MERA-2 data are available at https://doi.org/10.5067/5ESKGQTZG7FO, https://doi.org/10.5067/0JRLVL8YV2Y4 and https://doi. org/10.5067/V9208XZ30XBI. CFSR data are available at https://doi.org/10.1175/2010BAMS3001.1. Python scripts to estimate recycling following the EB94 method and the Brubaker method are available in an online repository: https://doi.org/10.5281/zenodo.6511636.



Acknowledgments

The research was supported by funding from the European Research Council under the European Union's Horizon 2020 research and innovation programme (DECAF project, Grant agreement no. 771492).

References

- Akinsanola, A. A., Ogunjobi, K. O., Abolude, A. T., & Salack, S. (2021). Projected changes in wind speed and wind energy potential over West Africa in CMIP6 models. *Environmental Research Letters*, *16*(4), 044033. https://doi.org/10.1088/1748-9326/abed7a
- Baker, J. C. A., Garcia-Carreras, L., Gloor, M., Marsham, J. H., Buermann, W., Da Rocha, H. R., et al. (2021). Evapotranspiration in the Amazon: Spatial patterns, seasonality, and recent trends in observations, reanalysis, and climate models. *Hydrology and Earth System Sciences*, 25(4), 2279–2300. https://doi.org/10.5194/hess-25-2279-2021
- Baker, J. C. A., & Spracklen, D. V. (2019). Climate benefits of intact Amazon forests and the biophysical consequences of disturbance. Frontiers in Forests and Global Change, 2. https://doi.org/10.3389/ffgc.2019.00047
- Bell, J. P., Tompkins, A. M., Bouka-Biona, C., & Sanda, I. S. (2015). A process-based investigation into the impact of the Congo basin deforestation on surface climate. *Journal of Geophysical Research: Atmospheres*, 120(12), 5721–5739. https://doi.org/10.1002/2014jd022586
- Bosilovich, M. G., & Chern, J.-D. (2006). Simulation of water sources and precipitation recycling for the MacKenzie, Mississippi, and Amazon River basins. *Journal of Hydrometeorology*, 7, 312–329. https://doi.org/10.1175/JHM501.1
- Bosilovich, M. G., & Schubert, S. D. (2002). Water vapor tracers as diagnostics of the regional hydrologic cycle. *Journal of Hydrometeorology*, 3(2), 149–165. https://doi.org/10.1175/1525-7541(2002)003<0149:wvtado>2.0.co;2
- Brubaker, K. L., Entekhabi, D., & Eagleson, P. S. (1993). Estimation of continental precipitation recycling. Journal of Climate, 6, 1077–1089. https://doi.org/10.1175/1520-0442(1993)006<1077:eocpr>2.0.co;2
- Burde, G. I., Gandush, C., & Bayarjargal, Y. (2006). Bulk recycling models with incomplete vertical mixing. Part II: Precipitation recycling in the Amazon basin. Journal of Climate, 19(8), 1473–1489. https://doi.org/10.1175/jcli3688.1
- Burnett, M. W., Quetin, G. R., & Konings, A. G. (2020). Data-driven estimates of evapotranspiration and its controls in the Congo Basin. Hydrology and Earth System Sciences, 24, 4189–4211.
- Butler, R. A. (2020). Deforestation: facts, figures, and pictures [Online]. Retrieved from https://rainforests.mongabay.com/deforestation/
- Cook, K. H., & Vizy, E. K. (2021). Hydrodynamics of regional and seasonal variations in Congo Basin precipitation. *Climate Dynamics*, 1–23. https://doi.org/10.1007/s00382-021-06066-3
- Costa, M. H., & Foley, J. A. (1999). Trends in the hydrologic cycle of the Amazon basin. Journal of Geophysical Research, 104(D12), 14189– 14198. https://doi.org/10.1029/1998jd200126
- Crowhurst, D., Dadson, S., Peng, J., & Washington, R. (2021). Contrasting controls on Congo Basin evaporation at the two rainfall peaks. *Climate Dynamics*, 56(5–6), 1609–1624. https://doi.org/10.1007/s00382-020-05547-1
- Crowhurst, D., Dadson, S. J., & Washington, R. (2020). Evaluation of evaporation climatology for the Congo basin wet seasons in 11 global climate models. *Journal of Geophysical Research: Atmospheres*, 125(6), e2019JD030619. https://doi.org/10.1029/2019jd030619
- da Rocha, H. R., Goulden, M. L., Miller, S. D., Menton, M. C., Pinto, L. D. V. O., De Freitas, H. C., & E Silva Figueira, A. M. (2004). Seasonality of water and heat fluxes over a tropical forest in eastern Amazonia. *Ecological Applications*, 14(sp4), 22–32. https://doi.org/10.1890/02-6001
- Davin, E. L., & de Noblet-Ducoudré, N. (2010). Climatic impact of global-scale deforestation: Radiative versus nonradiative processes. Journal of Climate, 23(1), 97–112. https://doi.org/10.1175/2009jcli3102.1
- Dominguez, F., Eiras-Barca, J., Yang, Z., Bock, D., Nieto, R., & Gimeno, L. (2022). Amazonian moisture recycling revisited using WRF with water vapor tracers. *Journal of Geophysical Research: Atmospheres*, 127(4), e2021JD035259. https://doi.org/10.1029/2021jd035259
- Dyer, E. L. E., Jones, D. B. A., Nusbaumer, J., Li, H., Collins, O., Vettoretti, G., & Noone, D. (2017). Congo Basin precipitation: Assessing seasonality, regional interactions, and sources of moisture. *Journal of Geophysical Research: Atmospheres*, 122(13), 6882–6898. https://doi. org/10.1002/2016id026240
- Eltahir, E. A., & Bras, R. L. (1994). Precipitation recycling in the Amazon basin. Quarterly Journal of the Royal Meteorological Society, 120(518), 861–880. https://doi.org/10.1002/qj.49712051806
- Eltahir, E. A., & Bras, R. L. (1996). Precipitation recycling. Reviews of Geophysics, 34(3), 367-378. https://doi.org/10.1029/96rg01927
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Flato, G. M. (2011). Earth system models: An overview. WIREs Climate Change, 2(6), 783-800. https://doi.org/10.1002/wcc.148
- Funk, C., Peterson, P., Landsfeld, M., Pedreros, D., Verdin, J., Shukla, S., et al. (2015). The climate hazards infrared precipitation with stations—A new environmental record for monitoring extremes. *Scientific Data*, 2(1), 150066. https://doi.org/10.1038/sdata.2015.66
- Gimeno, L., Stohl, A., Trigo, R. M., Dominguez, F., Yoshimura, K., Yu, L., et al. (2012). Oceanic and terrestrial sources of continental precipitation. *Reviews of Geophysics*, 50(4), RG4003. https://doi.org/10.1029/2012rg000389
- Haiden, T., Dahoui, M., Ingleby, B., De Rosnay, P., Prates, C., Kuscu, E., et al. (2018). Use of in situ surface observations at ECMWF. European Centre for Medium Range Weather Forecasts.
- Hansen, M. C., Potapov, P. V., Moore, R., Hancher, M., Turubanova, S. A., Tyukavina, A., et al. (2013). High-resolution global maps of 21st-century forest cover change. *Science*, 342(6160), 850–853. https://doi.org/10.1126/science.1244693
- Hari, V., Villarini, G., & Zhang, W. (2020). Fidelity of global climate models in representing the horizontal water vapour transport. *International Journal of Climatology*, 40(13), 5714–5726. https://doi.org/10.1002/joc.6546

Held, I. M., & Soden, B. J. (2006). Robust responses of the hydrological cycle to global warming. Journal of Climate, 19(21), 5686–5699. https:// doi.org/10.1175/jcli3990.1

- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Moriya, M., Onoda, H., et al. (2015). The JRA-55 reanalysis: General specifications and basic characteristics. 2, 93(1), 5–48. https://doi.org/10.2151/jmsj.2015-001
- Krasnopolsky, V. M., & Lin, Y. (2012). A neural network nonlinear multimodel ensemble to improve precipitation forecasts over continental US. Advances in Meteorology, 2012, 649450. https://doi.org/10.1155/2012/649450
- Lavers, D. A., Ralph, F. M., Waliser, D. E., Gershunov, A., & Dettinger, M. D. (2015). Climate change intensification of horizontal water vapor transport in CMIP5. *Geophysical Research Letters*, 42(13), 5617–5625. https://doi.org/10.1002/2015gl064672
- Lawrence, D., & Vandecar, K. (2015). Effects of tropical deforestation on climate and agriculture. *Nature Climate Change*, 5(1), 27–36. https://doi.org/10.1038/nclimate2430
- Lawrence, D., Hurtt, G. C., Arneth, A., Brovkin, V., Calvin, K. V., Jones, A. D., et al. (2016). The Land Use Model Intercomparison Project (LUMIP) contribution to CMIP6: Rationale and experimental design. *Geoscientific Model Development*, 9, 2973–2998. https://doi.org/10.5194/gmd-9-2973-2016

- Lettau, H., Lettau, K., & Molion, L. C. B. (1979). Amazonia's hydrologic cycle and the role of atmospheric recycling in assessing deforestation effects. *Monthly Weather Review*, *107*(3), 227–238. https://doi.org/10.1175/1520-0493(1979)107<0227:ahcatr>2.0.co;2
- Marengo, J. A. (2006). On the hydrological cycle of the Amazon basin: A historical review and current state-of-the-art. *Revista Brasileira de Meteorologia*, 21, 1–19.
- O'Neill, B. C., Kriegler, E., Ebi, K. L., Kemp-Benedict, E., Riahi, K., Rothman, D. S., et al. (2017). The roads ahead: Narratives for shared socioeconomic pathways describing world futures in the 21st century. *Global Environmental Change*, 42, 169–180. https://doi.org/10.1016/j. gloenvcha.2015.01.004
- Palmer, T. N., Shutts, G. J., Hagedorn, R., Doblas-Reyes, F. J., Jung, T., & Leutbecher, M. (2004). Representing model uncertainty in weather and climate prediction. Annual Review of Earth and Planetary Sciences, 33(1), 163–193. https://doi.org/10.1146/annurev.earth.33.092203.122552
- Perugini, L., Caporaso, L., Marconi, S., Cescatti, A., Quesada, B., de Noblet-Ducoudré, N., et al. (2017). Biophysical effects on temperature and precipitation due to land cover change. *Environmental Research Letters*, 12(5), 053002. https://doi.org/10.1088/1748-9326/aa6b3f
- Pokam, W. M., Djiotang, L. a. T., & Mkankam, F. K. (2012). Atmospheric water vapor transport and recycling in Equatorial Central Africa through NCEP/NCAR reanalysis data. *Climate Dynamics*, 38(9–10), 1715–1729. https://doi.org/10.1007/s00382-011-1242-7
- Potapov, P., Hansen, M. C., Laestadius, L., Turubanova, S., Yaroshenko, A., Thies, C., et al. (2017). The last frontiers of wilderness: Tracking loss of intact forest landscapes from 2000 to 2013. *Science Advances*, 3(1), e1600821. https://doi.org/10.1126/sciadv.1600821
- Ralph, F. M., Neiman, P. J., & Rotunno, R. (2005). Dropsonde observations in low-level jets over the northeastern Pacific Ocean from CALJET-1998 and PACJET-2001: Mean vertical-profile and atmospheric-river characteristics. *Monthly Weather Review*, 133(4), 889–910. https://doi. org/10.1175/mwr2896.1
- Richter, I., & Tokinaga, H. (2020). An overview of the performance of CMIP6 models in the tropical Atlantic: Mean state, variability, and remote impacts. *Climate Dynamics*, 55(9–10), 2579–2601. https://doi.org/10.1007/s00382-020-05409-w
- Richter, I., Xie, S.-P., Wittenberg, A. T., & Masumoto, Y. (2012). Tropical Atlantic biases and their relation to surface wind stress and terrestrial precipitation. *Climate Dynamics*, 38(5–6), 985–1001. https://doi.org/10.1007/s00382-011-1038-9
- Saatchi, S. S., Harris, N. L., Brown, S., Lefsky, M., Mitchard, E. T. A., Salas, W., et al. (2011). Benchmark map of forest carbon stocks in tropical regions across three continents. *Proceedings of the National Academy of Sciences*, 108(24), 9899–9904. https://doi.org/10.1073/ pnas.1019576108
- Salati, E., Dall'olio, A., Matsui, E., & Gat, J. R. (1979). Recycling of water in the Amazon basin: An isotopic study. Water Resources Research, 15(5), 1250–1258. https://doi.org/10.1029/wr015i005p01250
- Skliris, N., Zika, J. D., Nurser, G., Josey, S. A., & Marsh, R. (2016). Global water cycle amplifying at less than the Clausius-Clapeyron rate. Scientific Reports, 6(1), 38752. https://doi.org/10.1038/srep38752
- Sorí, R., Nieto, R., Vicente-Serrano, S. M., Drumond, A., & Gimeno, L. (2017). A Lagrangian perspective of the hydrological cycle in the Congo River basin. *Earth Syst. Dynam.*, 8(3), 653–675. https://doi.org/10.5194/esd-8-653-2017
- Spracklen, D. V., Arnold, S. R., & Taylor, C. M. (2012). Observations of increased tropical rainfall preceded by air passage over forests. *Nature*, 489(7415), 282–285. https://doi.org/10.1038/nature11390
- Spracklen, D. V., Baker, J. C. A., Garcia-Carreras, L., & Marsham, J. (2018). The effects of tropical vegetation on rainfall. Annual Review of Environment and Resources, 43, 14.1–14.26. https://doi.org/10.1146/annurev-environ-102017-030136
- Spracklen, D. V., & Garcia-Carreras, L. (2015). The impact of Amazonian deforestation on Amazon basin rainfall. *Geophysical Research Letters*, 42(21), 9546–9552. https://doi.org/10.1002/2015gl066063
- Staal, A., Tuinenburg, O. A., Bosmans, J. H. C., Holmgren, M., Van Nes, E. H., Scheffer, M., et al. (2018). Forest-rainfall cascades buffer against drought across the Amazon. *Nature Climate Change*, 8(6), 539–543. https://doi.org/10.1038/s41558-018-0177-y
- Trenberth, K. E. (1999). Atmospheric moisture recycling: Role of advection and local evaporation. Journal of Climate, 12(5), 1368–1381. https://doi.org/10.1175/1520-0442(1999)012<1368:amrroa>2.0.co;2
- Tuinenburg, O. A., Theeuwen, J. J. E., & Staal, A. (2020). High-resolution global atmospheric moisture connections from evaporation to precipitation. Earth System Science Data, 12(4), 3177–3188. https://doi.org/10.5194/essd-12-3177-2020
- Tyukavina, A., Hansen, M. C., Potapov, P., Parker, D., Okpa, C., Stehman, S. V., et al. (2018). Congo Basin forest loss dominated by increasing smallholder clearing. *Science Advances*, 4(11), eaat2993. https://doi.org/10.1126/sciady.aat2993
- van der Ent, R. J., Savenije, H. H., Schaefli, B., & Steele-Dunne, S. C. (2010). Origin and fate of atmospheric moisture over continents. Water Resources Research, 46(9), 1–12. https://doi.org/10.1029/2010wr009127
- von Randow, C., Manzi, A. O., Kruijt, B., De Oliveira, P. J., Zanchi, F. B., Silva, R. L., et al. (2004). Comparative measurements and seasonal variations in energy and carbon exchange over forest and pasture in South West Amazonia. *Theoretical and Applied Climatology*, 78(1–3), 5–26. https://doi.org/10.1007/s00704-004-0041-z
- Yang, Z., & Dominguez, F. (2019). Investigating land surface effects on the moisture transport over South America with a moisture tagging model. *Journal of Climate*, 32(19), 6627–6644. https://doi.org/10.1175/jcli-d-18-0700.1
- Zemp, D., Schleussner, C.-F., Barbosa, H., Van Der Ent, R., Donges, J. F., Heinke, J., et al. (2014). On the importance of cascading moisture recycling in South America. Atmospheric Chemistry and Physics, 14(23), 13337–13359. https://doi.org/10.5194/acp-14-13337-2014

References From the Supporting Information

Brubaker, K. L., Entekhabi, D., & Eagleson, P. S. (1994). Atmospheric water vapor transport and continental hydrology over the Americas. *Journal of Hydrology*, 155(3–4), 407–428. https://doi.org/10.1016/0022-1694(94)90180-5

Budyko, M. (1974). Climate and life. Academic Press.

- Builes-Jaramillo, A., & Poveda, G. (2018). Conjoint analysis of surface and atmospheric water balances in the Andes-Amazon system. Water Resources Research, 54(5), 3472–3489. https://doi.org/10.1029/2017wr021338
- Burde, G. I., & Zangvil, A. (2001a). The estimation of regional precipitation recycling. Part I: Review of recycling models. *Journal of Climate*, 14(12), 2497–2508. https://doi.org/10.1175/1520-0442(2001)014<2497:teorpr>2.0.co;2
- Burde, G. I., & Zangvil, A. (2001b). The estimation of regional precipitation recycling. Part II: A new recycling model. *Journal of Climate*, 14(12), 2509–2527. https://doi.org/10.1175/1520-0442(2001)014<2509:teorpr>2.0.co;2
- Gelaro, R., Mccarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., et al. (2017). The modern-era retrospective analysis for research and applications, version 2 (MERRA-2). Journal of Climate, 30(14), 5419–5454. https://doi.org/10.1175/jcli-d-16-0758.1
- Global Modeling and Assimilation Office (GMAO). (2015a). MERRA-2 instM_2d_asm_Nx: 2d,Monthly mean,Single-Level,Assimilation,Single-level diagnostics V5.12.4. Goddard Earth Sciences Data and Information Services Center (GES DISC).



- Global Modeling and Assimilation Office (GMAO). (2015b). MERRA-2 instM_3d_ana_Np: 3d,Monthly mean,Instantaneous,Pressure-Level,Analysis,Analyzed meteorological fields V5.12.4. Goddard Earth Sciences Data and Information Services Center (GES DISC).
 Global Modeling and Assimilation Office (GMAO). (2015c). MERRA-2 targM_2d_flx_Nx: 2d,Monthly mean,Time-Averaged,Single-Level,As-
- similation, Surface flux diagnostics V5.12.4. Goddard Earth Sciences Data and Information Services Center (GES DISC).

Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., et al. (2019a). *ERA5 monthly averaged data on pressure levels from 1979 to present*, Copernicus Climate Change Service (C3S) Climate Data Store (CDS).

Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., et al. (2019b). *ERA5 monthly averaged data on single levels from 1979 to present*, Copernicus Climate Change Service (C3S) Climate Data Store (CDS).

Japan Meteorological Agency/Japan. (2013). JRA-55: Japanese 55-year reanalysis, monthly means and variances. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory.

Nicholson, S. E., Klotter, D., Zhou, L., & Hua, W. (2019). Validation of satellite precipitation estimates over the Congo Basin. Journal of Hydrometeorology, 20(4), 631–656. https://doi.org/10.1175/jhm-d-18-0118.1

- Peixóto, J. P., & Oort, A. H. (1984). Physics of climate. Reviews of Modern Physics, 56(3), 365–429. https://doi.org/10.1103/revmodphys.56.365 Saha, S., Moorthi, S., Pan, H.-L., Wu, X., Wang, J., Nadiga, S., et al. (2010a). The NCEP climate forecast system reanalysis. Bulletin of the American Meteorological Society, 91(8), 1015–1058. https://doi.org/10.1175/2010bams3001.1
- Saha, S., Moorthi, S., Pan, H.-L., Wu, X., Wang, J., Nadiga, S., et al. (2010b). NCEP climate forecast system reanalysis (CFSR) monthly products, January 1979 to December 2010. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory.

Swann, A. L. S., & Koven, C. D. (2017). A direct estimate of the seasonal cycle of evapotranspiration over the Amazon basin. Journal of Hydrometeorology, 18(8), 2173–2185. https://doi.org/10.1175/jhm-d-17-0004.1